ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Neumann, Anne; Sorge, Lars; von Hirschhausen, Christian; Wealer, Ben

Article — Accepted Manuscript (Postprint) Democratic quality and nuclear power: Reviewing the global determinants for the introduction of nuclear energy in 166 countries

Energy Research & Social Science

Provided in Cooperation with: German Institute for Economic Research (DIW Berlin)

Suggested Citation: Neumann, Anne; Sorge, Lars; von Hirschhausen, Christian; Wealer, Ben (2020) : Democratic quality and nuclear power: Reviewing the global determinants for the introduction of nuclear energy in 166 countries, Energy Research & Social Science, ISSN 2214-6296, Elsevier BV, Amsterdam, Vol. 63, https://doi.org/10.1016/j.erss.2019.101389 , https://www.sciencedirect.com/science/article/abs/pii/S221462961930667X?via%3Dihub

This Version is available at: https://hdl.handle.net/10419/266405

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



ND http://creativecommons.org/licenses/by-nc-nd/4.0





Democratic Quality, Development, and Nuclear Warheads: Reviewing the Global Determinants for the Introduction of Nuclear Power in 166 Countries

Anne Neumann^{*ac}, Lars Sorge^{†bc}, Christian von Hirschhausen^{‡bc}, and Ben Wealer^{§bc}

^aNTNU Trondheim, ^bDIW Berlin, ^cBerlin University of Technology

November 2019

Abstract: This paper analyzes the nature of democratic development in a nation on the process of introducing nuclear power over the period 1960 - 2017 for an unbalanced panel of 166 countries. Given the involved political process of introducing nuclear power and its political importance, as well as proposals to construct new nuclear reactors in currently about 30 countries, this question is both of historic and current interest. We apply a multinomial logistic regression approach that relates the likelihood of a country to introduce nuclear power to its level of democratic quality and nuclear warhead possession. The model results suggest that countries with lower levels of democratic development are more likely to introduce nuclear power. Our results moreover indicate that countries which possess at least one nuclear warhead are more likely to continue to use nuclear power instead of not using nuclear power at all. We discuss these results in the context of the public policy debate on nuclear power, yet beyond energy and environmental issues addressing the neglected political and democratic dimension in connection with nuclear power.

Keywords: Nuclear power, nuclear energy, nuclear warhead, democracy, multinomial logit, panel data

© <2021>. This manuscript version is made available under the CC-BY-NC-ND 4.0 license http://creativecommons.org/licenses/by-nc-nd/4.0/

^{*}anne.neumann@ntnu.no

[†]Corresponding author: lsorge@diw.de

[‡]cvh@wip.tu-berlin.de

[§]bw@wip.tu-berlin.de

An earlier version of this paper was presented at the Low Carbon Transformation and Sustainable Development - Status Quo and Research Outlook in Berlin in April 2018, at the 41^{st} IAEE International Conference in Groningen in June 2018, at the Berlin Conference on Sustainable Energy and Infrastructure Economics and Policy (BELEC) in Berlin in October 2018, and at the Bavarian Berlin Energy Research (BB²) Workshop in Munich in February 2019. We thank the seminar and conference participants for fruitful discussion and input. We also thank the editor and three anonymous referees that helped to improve the quality of the paper. All remaining errors are ours. Declarations of interest: none.

1 Introduction

Nuclear power was one of the most important developments of the 20th century, and it continues to affect discussions about energy security, climate change, and geopolitics well into the 21st century. Nuclear power emerged from the combination of "basic science and warfare" [1] in the 1940s. Decisions in this sector have always been based on political bargaining and state financing, rather than on pure economic rationality [2–4]. Understanding the drivers of national decisions to "go nuclear", i.e. to bring nuclear power plants online in a country, is therefore crucial not only for interpreting the history, but also the future perspectives of nuclear technology: At the time of this paper, 31 countries depend on nuclear power to produce electricity, and approximately thirty more are debating, planning, or building nuclear power generation [5]. The International Atomic Energy Agency (IAEA) still considers a high global potential for nuclear power up to the year 2050 [6].

The political nature of decisions on nuclear power raise interesting questions, in particular with respect to the drivers of these decisions, the criteria for going nuclear, (or not going nuclear), and the decision-making and implementation process. The institutional and political framework of the participating actors moreover plays an important role in particular against the background of higher overnight costs of nuclear power compared to coal and natural gas [7,8]. The technical complexity of nuclear power and the need for strong vertical and horizontal coordination within the sector suggests centralized decision-making, in addition to political, cultural, and social characteristics of a country that influence nuclear trajectories [9]. In that context, the nature of the political system, e.g. the degree of democratic and competitive decision-making, can be expected to be an important variable. Socio-cultural, political and economic conditions which encourage the deployment of nuclear power, have already been acknowledged in qualitative multi-country case studies (e.g., [10, 11]), yet empirical research emphasizing in particular the political economy dimension of nuclear power is surprisingly scarce.

To fill this research gap, this paper empirically analyzes the relationship between between a country's decision to introduce nuclear power - defined as connecting the first nuclear power plant to the grid - and the level of democratic development. We use observable characteristics from the Power Reactor Information System (PRIS) database [12] to specify three categories which distinctively define a countries' nuclear energy strategy chosen over time. At each point in time we evaluate if countries currently use, have used at any point in time, or start to construct a nuclear power plant. Any given country which has no nuclear power plant under construction or operational at time t is categorized as "non-nuclear". We define the period of the construction start of the first nuclear reactor until the first grid connection of any nuclear power plant as the observable outcome to "go nuclear", which then represents the introduction of nuclear power. At any point in time countries are categorized as "nuclear" if they have at least one nuclear power reactor fully operationally at time t. The three distinct observable outcomes for a countries' nuclear energy strategy are operationalized as our dependent variable which is categorical and of unordered nature. We use our main predictor, the level of democratic development in both continuous and categorical modes, to measure the degree of institutionalized democracy and autocracy, respectively. We include national development, energy transitions, and environmental indicators as identified in the published literature as the explanatory variables and also to control for nuclear warhead possession.

Our initial hypothesis, based on the existing literature and the qualitative case studies from Sovacool and Valentine [10] and Valentine and Sovacool [11], is that due to the complex, and often controversial, political decisions required to develop nuclear power, less democratically governed countries are more likely to enter the sector and introduce nuclear power than countries with higher levels of democracy. We analyze the impacts of democratic development on the introduction of nuclear power from 1960 to 2017 for an unbalanced panel of 166 countries. Different from the previous literature (e.g., [13]), we focus on the motives affecting the initial introduction of nuclear power emphasizing the level of democratic development as a key determinant for the nuclear energy choice while covering a broader time frame than previously examined. Our model is based on a multinomial logistic regression approach that relates the likelihood of a country to introduce nuclear power to its level of democratic development. The model results robustly suggest that countries with lower levels of democratic development are more likely to introduce nuclear power. Our empirical analysis thus provides a robust statistical assessment of the democracy and nuclear power nexus and identifies the variable democratic quality as an important factor to the nuclear energy choice. We contribute to the ongoing discussion about nuclear energy by shifting the focus beyond economic, climate, and environmental issues towards the frequently neglected political and democratic dimension of nuclear power. According to Gralla et al. [9], focusing on the political dimension of nuclear power is highly relevant in order to complement the global view on nuclear energy.

The remainder of this paper is organized as follows. Section 2 provides the background and reviews the empirical literature in connection with nuclear power and democracy. Section 3 presents the data and explains our empirical approach. Section 4 reports the empirical results. Section 5 presents the checks for robustness. Section 6 concludes with a discussion and offers suggestions for future research.

2 Background and Related Literature

This section draws on both theoretical and empirical literature to present an overview of which factors out of the political realm in particular are considered to be relevant for nuclear power development. Section 2.1 presents the theoretical underpinnings of our empirical analysis and Section 2.2 reviews related empirical applications. In general, the early literature on nuclear power uses detailed case studies to analyze how nuclear power developed in different political contexts. A more empirically orientated strand analyzes aggregate indicators which might facilitate the development of nuclear power development within a cross-country set up [14]. However, both, empirical analyses investigating the relationship between nuclear power deployment and democracy in particular, and contributions concerning more generally socioeconomic factors of nuclear power deployment are surprisingly scarce.

2.1 Theoretical Underpinnings

The multi-level perspective (MLP) is a useful analytical tool to determine how a country's level of democratic development may affect the introduction of nuclear power. The MLP understands transitions as outcomes of alignment between developments at multiple levels. It consists of the three analytical levels niche-innovations, sociotechnical regimes, and sociotechnical landscapes which help explain transitions based on interactions between processes at these three levels [15]. The sociotechnical landscapes at the top level is an exogenous environment beyond the direct influence of actors within sociotechnical regimes and niche-innovations [15].

The sociotechnical regime layer contains multiple dimensions such as policy, technology, user practices, science, cultural meaning, infrastructure and industry [16]. The energy supply sector is embedded in the sociotechnical regime layer of this multi-level environment and can be conceptualized as a sociotechnical system which similarly consists of actors, institutions, as well as material artifacts and knowledge. The diverse components of the system interact, are interrelated and dependent on each other [17]. Nuclear power thus is situated in a social and political environment that influences its evolution and interacts with different socio-economic institutional settings and various stakeholders during development, construction, and operation. In this regard, large scale energy technologies are considered not as a determinant of political regimes but rather as co-evolving with a country's socio-economic institutions, actors, and social norms over the operational lifetime which usually spans over decades.¹ Conventional nuclear power plants - unlike distributed renewable energy infrastructures such as wind and solar - generate electricity on a massive scale in a centralized location. Nuclear energy consequently has

¹In the United States, some aging reactors have received lifetime extensions to 80 years. If decommissioning and long-term storage of radioactive waste management also is considered, a plant's lifetime extends to a million years.

found its particular niche in providing baseload electricity in the energy system [18,19]. The introduction of nuclear energy thus depends on the relationships between the three layers and the involvement of a variety of factors including policy, technology, user practices, science, cultural meaning, infrastructure, and industry. The dynamics of sociotechnical systems associated with nuclear power are related to the general political culture, elite policy discourse, patterns of public opinion and wider attributes of democratic governance which represent general qualities of democracy [20].

Democratic quality is multidimensional in nature with different aspects of democratic quality overlapping. Diamond and Morlino [21] identify eight dimensions: the rule of law; participation; competition; vertical accountability; horizontal accountability; respect for civil and political freedoms; progressive implementation of greater political (and underlying it, social and economic) equality; and responsiveness on which democracies diverge in terms of quality. The linkages among the different aspects of democracy "[...] interact and reinforce one another, ultimately converging into a system". Diamond and Morlino [21] moreover define minimum standards for democracy: universal, adult suffrage; recurring, free, competitive, and fair elections; more than one serious political party; and alternative sources of information.

There are no universal definitions, however, of democracy and democratic quality. Zakaria [22] defines a liberal democracy as a political system characterized by free and fair elections, the rule of law, separation of powers, and the protection of basic liberties of speech, assembly, religion, and property. The author notes that few countries characterized as falling between democratic and nondemocratic matured into liberal democracies during the 1990s. Zakaria [22] coined the term "illiberal democracy" to describe countries that have free and fair elections but "are routinely ignoring constitutional limits on their power and depriving their citizens of basic rights and freedoms". In such countries, "[a] weak rule of law will likely mean that participation by the poor and marginalized is suppressed, individual freedoms are insecure, many civic groups are unable to organize and advocate, the resourceful and well-connected are unduly favored, corruption and abuse of power run rampant, political competition is unfair, voters have a hard time holding rulers to account, and overall democratic responsiveness is gravely enfeebled" [21].

Another analytical concept related to our analysis is the framework of concentrated energy-politics vs. distributed energy-politics: the spatial distribution of energy infrastructures influences democratic development and the degree of democratic development influences the spatial distribution of energy infrastructures. According to Burke and Stephens [23], due to their inherent flexibility decentralized energy technologies are considered to more readily organize and enable distributed political and economic power, and vice versa. This relationship is characterized as strongly democratic and described as distributed energy-politics. On the contrary, energy systems based on concentrated energy sources are considered to organize and enable more concentrated forms of power and centralized or authoritarian political relationships, and vice versa. This relationship thus is characterized as weakly democratic and refers to concentrated energy-politics. To what extent political power is concentrated and democracy is developed may impact the deployment of certain energy infrastructures, but the developed energy system similarly may influence the level of democratic development. With respect to nuclear power, Bookchin [24] argues that the enhancement of democracy by decentralizing power is prevented by the continuation of nuclear energy. Lovins [25] similarly questions the democratic extension capabilities of nuclear energy as it is a "hard" centralized energy path which in part due to its inherent nuclear weapon proliferation potential affects society in terms of authoritarian forms of governance [20].

Historically, certain environments and conditions have encouraged the development of nuclear power. Sovacool and Valentine [10] and Valentine and Sovacool [11] develop a theoretical framework consisting of six influential factors: strong state involvement in guiding economic development; centralization of national energy planning; campaigns linking technological progress to national revitalization; influence of technocratic ideology on policy decisions; subordination of challenges to political authority; and low levels of civic activism.²

Strong centrally led economic planning and state involvement either directly through government action or indirectly through state-owned utilities is considered necessary due to the inflexibility and complexity of nuclear power and a high degree of supply chain coordination to realize such energy mega-projects. Similarly, centralization of energy planning facilitates the necessary control and enables to overcome disagreements internally which lowers transaction costs during resolution processes encouraging the expansion of nuclear power. Since nuclear power historically is associated with technological progress and modernity, governmental strategies committed to link technological developments to a national renaissance encourages a national culture which is more likely to tolerate the risks associated with nuclear power. When technocratic ideology strongly influences public policy, the necessary ideological support for nuclear power development is provided. Conditions under which political and public debate are minimized more easily enable the implementation of governmental programs which run contrary to public interest. Lastly, environments which eliminate civic activism detrimentally impact public opposition which could oppose the development of nuclear power programmes. According to Sovacool and Valentine [10] and Valentine and Sovacool [11] these six catalysts are simultaneously political, social, and economic. However, only the political environment can influence and overpower both the social and economic dimension at least in authoritarian regimes more easily. Influential factors such as strong state involvement in economic development, centralization of national energy planning, subordination of challenges to political authority, and low levels of civic activism concern the realms of democracy are in connection with rather authoritarian than democratic regime characteristics.³

 $^{^{2}}$ Sovacool and Valentine [10] moreover identified the abatement of greenhouse gas emissions as a potential seventh factor emerging in the environmental policy realm.

³Military rule in South Korea for instance allowed the government to control the policy agenda and strong autocratic control over the economy and society until the late 1980s. In Japan, government control over the media in the 1960s and 1970s diluted political and popular opposition to nuclear power development and the dangers. In China, although environmental activists collected about one million signatures against the Daya Bay nuclear power plant, the government detained and arrested protesters ([10,11]).

Kitschelt [26], who compares anti-nuclear protest movements in France, Sweden, the United States, and West Germany, argues that a country's political and institutional dimensions in which social movements operate shape the level and pattern the protests. Mobilization strategies and impacts of social movements can be explained partly by the general characteristics of political opportunity structures. In other words, the chances of broad mobilization increase when anti-nuclear movements can easily collect and disseminate information which in turn can influence policies concerning nuclear power development and expansion. Already Weinberg [27] recognizes the importance of public perception towards nuclear power: "The public perception and acceptance of nuclear energy [...] has emerged as the most critical question concerning the future of nuclear energy". O'neil [28], who analyzes the development of nuclear energy in transformation states, argues that citizens' willingness to protest may be influenced by the country's level of nuclear energy dependence, i.e., anti-nuclear movements are less likely to occur in countries with a relatively large dependence on nuclear energy due to the public's fear of higher electricity bills or fear of negative growth effects. O'neil [28] also concludes that support or opposition to nuclear power is "not a function of democracy but rather of complex relationships between state, society, and the institutions they create".

Jewell and Ates [14] emphasize the importance of political stability for both internal (program constancy and reliability) and external (investor confidence) support of nuclear power development. Jewell [29] also observes that in politically unstable countries, the introduction of nuclear power for civil purposes in conjunction with developing nuclear weapons is only possible by mobilizing extraordinary political will and resources. This logic in particular applies to countries such as India and Pakistan which both are characterized by low levels of the World Bank Political Stability Index (PSI). Both countries have experienced political instability in the years preceding or following construction of their first nuclear power plants according to the Political Instability Task Force (PITF), which records events in connection with the occurrence of "partial or total state failure".

Following World War II, nuclear power first emerges as the "child of science and warfare" [1] in the victorious countries of the USA, the USSR, the UK, and France, and

later in China. Today, nuclear power sits at the intersection of military use and electricity generation. Over time, the development of atomic energy for civil purposes and for military use has become interchangeable and interdependent [30].

In fact, Hirschhausen [2] argues that nuclear power has to be analyzed under the topic of joint production (so called "economies of scope") as nuclear co-production includes military goods (e.g. plutonium, tritium) as well as civilian goods and services (e.g. electricity, medical services). Related to this, Stirling and Johnstone [3] emphasize the importance of industrial supply chains involving the wider nuclear skills, education, research, design, engineering, and industrial capabilities necessary to sustain or introduce nuclear weapon programmes as well as for nuclear powered submarines capabilities.

2.2 Related Empirical Work

Fuhrmann [13] uses a probit model to empirically identify factors which encourage 129 countries to build nuclear power plants from 1965 to 2000. Based on information from the International Atomic Energy Agency's Power Reactor Information System (PRIS) database, the dependent variable is dichotomous and coded as 1 if a country begins building a reactor in year t+1, and 0 otherwise. As predictor variables, Fuhrmann [13] includes GDP as a proxy for economic capacity, energy dependence, an indicator for nuclear weapons exploration, a dummy variable which indicates if a state shares a defense pact with a major supplier of nuclear power plants, Nuclear Non-Proliferation Treaty (NPT) membership dummies, and nuclear accidents dummies which interact with the composite indicator from the Polity IV Project measuring a country's regime type. The empirical results indicate that higher levels of economic development are associated with a higher probability for construction and that countries which become less dependent on energy imports are less likely to build nuclear reactors. The indicator for nuclear weapons exploration, the supplier alliance dummy, and the NPT indicator are statistically insignificant. Fuhrmann [13] also shows that the impacts of nuclear accidents on construction depend on regime type, i.e., highly authoritarian states tended to be less affected by the Chernobyl disaster than countries with high levels of democratic development.

Yamamura [31], who empirically analyzes the effect of free media on the Japanese public's view of nuclear energy after the 2011 Fukushima Daiichi nuclear disaster, uses cross-sectional panel data of 37 countries collected approximately two weeks after the disaster. From the survey the author obtains the rate of agreement that nuclear power plants are properly secured against accidents and uses it as the dependent variable, controlling for the presence of nuclear power plants, freedom of expression and media, total number of natural disasters since 1970, GDP per capita, government expenditures, and including dummies for East Asian countries. The results show that freedom of expression and media significantly influences views on the security of nuclear power plants. Citizens tend to disagree that nuclear power plants are properly secured against accidents when the political setting assures both freedom of expression and media to a greater extent. The results moreover show that freedom of media leads citizens to support the presence of nuclear energy. The latter result seems to contradict our reasoning above. It however rather highlights the importance of freedom of media to be guaranteed. Only when freedom of media is guaranteed, citizens are able to evaluate costs and benefits associated with the presence of nuclear energy to then decide informed about nuclear policy [31].

Gralla et al. [9] group countries according to the nuclear energy strategies (no nuclear production, phase-out, planning to produce, produce nuclear energy) of the World Nuclear Association. Based on the statistical mean for the respective group on 20 indicators from 1960 to 2013, nuclear countries have higher per capita energy use, carbon dioxide emissions, and household final consumption expenditures compared to countries planning to use nuclear energy and countries without nuclear energy use. Gralla et al. [9] use a generalized linear mixed model (GLMM) and the nuclear energy status of all countries between 1960 and 2013 as the dependent variable to identify the socioeconomic, technological, and environmental indicators correlating with the starting year of each country's nuclear energy production. They find that 28 out of the 96 world development indicators significantly correlate with the start of nuclear energy production. Gralla et al. [9], however, do not control for the level of democratic or institutional development.

3 Data and Methodology

Section 3.1 describes the three categories which define a country's nuclear energy strategy chosen over time and presents a descriptive analysis of the utilized panel dataset. Section 3.2 discusses the methodology for predicting the probability of category membership in order to analyze if democracies tend not to start constructing nuclear power plants compared to democratically less developed countries.

3.1 Data

Our analysis covers the period from 1960 to 2017 and we construct an unbalanced panel time series dataset covering 166 countries.⁴ We empirically analyze how the level of democratic development impacts a countries' choice to introduce nuclear power, controlling for nuclear warhead ownership, national development, and both energy transitions as well as environmental indicators.

Our polytomous dependent variable is based on information from the Power Reactor Information System (PRIS) database [12] and contains the three categories "non-nuclear", to "go nuclear", and "nuclear" which represent the current nuclear energy statuses of a given country in a given year to establish a countries' nuclear energy strategy chosen over

⁴The World Development Indicators (WDI) database from The World Bank originally includes 217 countries. The Polity IV Project data set includes only 167 countries. Therefore, we exclude the 50 countries not included in the Polity IV Project data set and also Taiwan, which is not covered in the WDI database, to obtain 166 countries. We exclude: American Samoa, Andorra, Antigua and Barbuda, Aruba, Bahamas The, Barbados, Belize, Bermuda, British Virgin Islands, Brunei Darussalam, Cayman Islands, Channel Islands, Curacao, Dominica, Faroe Islands, French Polynesia, Gibraltar, Greenland, Grenada, Guam, Hong Kong SAR China, Iceland, Isle of Man, Kiribati, Liechtenstein, Macao SAR China, Maldives, Malta, Marshall Islands, Micronesia Fed. Sts., Monaco, Nauru, New Caledonia, Northern Mariana Islands, Palau, Puerto Rico, Samoa, San Marino, Sao Tome and Principe, Seychelles, St. Maarten (Dutch part), St. Kitts and Nevis, St. Lucia, St. Martin (French part), St. Vincent and the Grenadines, Tonga, Turks and Caicos Islands, Tuvalu, Vanuatu, US Virgin Islands, and West Bank and Gaza.

time. The unit of analysis thus is the country-year. The country-years take on the value zero if the respective country has no nuclear power plant under construction or operational in any given year. The period of the construction start of the first nuclear reactor until the first grid connection of any nuclear power plant is coded as one for each country-year which represents the introduction of nuclear power. Accordingly, this indicates the period of the completion of the first nuclear reactor project. Nuclear country-years are coded as two if at least one nuclear power reactor is fully operationally.⁵

Our key predictor of interest, the level of democratic development, comes from the Polity IV Project from the Center for Systemic Peace. The index from the Polity IV Project is a combination of the institutionalized democracy and autocracy indicator. The Polity score is computed by subtracting the autocracy from the democracy score which results in an unified polity scale ranging from +10 (strongly democratic) to -10 (strongly autocratic). We use the Polity2 score which is a modified version of the Polity index to facilitate the use in time-series analyses [32]. Following Haber and Menaldo [33] we first normalize the Polity2 index to run from 0 to 100 to obtain a continuous democracy variable D1. To obtain a categorical measure for the democracy levels, we classify countries with a score of D1 > 66 as democratically free F, 33 < D1 < 66 as democratically partly free PF, and D1 < 33 as democratically not free NF to operationalize fully liberal

⁵To define a country's nuclear energy status, we follow the literature, e.g. Jewell and Ates [14]; the International Atomic Energy Agency (IAEA) [34] uses the period after the grid connection date of a nuclear reactor as the definition. We also code the introduction of nuclear power for either the period of the construction start until the respective reactor reaches first criticality as well as until the commercial operation date. Our results, however, do not change for choosing the period until first criticality as well as until the commercial operation date (we thank an anonymous referee for raising this important issue). We moreover note that Italy started with the construction of the first nuclear reactor on 01 November 1958 which was connected to the grid on 12 May 1963. The last reactor in Italy was shut down on 01 July 1990. Kazakhstan started the construction of the first nuclear reactor on 01 October 1964 which was connected to the grid on 16 July 1973. The reactor was shut down on 22 April 1999. Lithuania started with the construction of 10 May 1977 which was connected to the grid on 31 December 1983. The last reactor in Lithuania however was shut down on 31 December 2009.

democratic characteristics, illiberal democratic characteristics, and non-democratic regime characteristics, respectively. 6

The decision of a country to introduce nuclear power might partly be driven by the aim to develop nuclear weapon programmes. Nuclear reactors fueled by uranium used for generating civilian electric power accumulate plutonium. Nuclear power producing countries over time acquire enough quantities of plutonium usable for nuclear weapons [35]. It is therefore only a question of political will and willingness to develop nuclear weapons or not for nuclear power producing countries [36]. We construct an indicator W which takes on the value one if a country possesses at least one nuclear warhead in a given year and zero otherwise.⁷ Moreover, we control for national development (GDP per capita, urbanization) and both energy transitions and environmental indicators (electric power consumption, fossil fuel rents, energy imports, CO₂ emissions per capita) from the World Development Indicators (WDI) database from The World Bank which all significantly correlate with nuclear energy production [9].

GDP per capita GDP is measured in constant 2010 USD and our main indicator for a countries' financial capacity. A high degree of national financial capacity for nuclear power development is necessary to allocate initial investments for creating the regulatory, legislative and basic physical infrastructure before construction, but similarly required

⁶We use the democracy indicator from the Polity IV Project instead of using data from Freedom House, another commonly used indicator for democratic quality, because the indicator from Freedom House is available only from 1972 on; 1989 data are missing; and Argentina, Armenia, Belgium, Brazil, Bulgaria, Canada, Finland, Germany, India, Italy, Japan, Kazakhstan, Korea, Rep., Netherlands, Pakistan, Slovak Republic, Spain, Sweden, Switzerland, and Ukraine all started constructing their first reactors before 1972. Put differently, if we use the indicator Freedom House, we effectively loose information on 20 out of 38 (53%) countries which ever have build a nuclear power plant. A detailed description of the underlying methodology from the Polity IV Project is available at https://www.systemicpeace.org/. We moreover note that the various indicators for democracy such as the Freedom in the World rating from Freedom House, the Democracy Ranking by the Democracy Ranking Association, the Economist Intelligence Unit's Democracy Index, or the Democracy Barometer for instance vary in terms of their methodology and how democratic quality in particular is ranked and assessed. Democracy thus is a highly contested concept and measured in various ways as pointed out by two anonymous referees.

⁷The dates for nuclear warhead possession are based on Kristensen and Norris [37]. South Africa initially is not coded as a nuclear weapons states due to the lack of comparable information to Kristensen and Norris [37]. According to Jo and Garzke [38], the entire period of proliferation in South Africa dates from 1979 to 1991. Thus, we would have had to compare the period of a nuclear weapons program with the actual possession of at least one nuclear warhead. Our results, however, do not change if we code South Africa as a nuclear weapons state over the respective period. We thank an anonymous referee for raising this important issue.

to finance actual construction of the first nuclear power plant [29]. Urbanization U is measured as the share of the population living in urban areas and reflects the transition from rural to urban areas. Urbanization intensifies the demand for urban infrastructure and transportation, and stimulates the concentration of consumption and production which is associated with increasing energy demand [39]. To control for a countries' electricity demand, we use electric power consumption E measured in kWh per capita as an additional predictor variable. Energy security considerations can translate into motivations for pursuing nuclear energy in order to increase energy independence. In countries such as Japan, UK, France, and Finland, independence of energy imports are main arguments for supporting nuclear power [9,29]. We use energy imports EI (% of primary energy use) to measure energy security and independence. In countries which are richly endowed with fossil fuels, the presence of cheap and abundant domestic fuels is expected to similarly affect a countries' energy mix and thus the likelihood for nuclear power deployment. We thus construct an indicator for fossil fuel rents FFR which are the sum of oil rents, natural gas rents, coal rents measured in percentage of GDP. CO₂ emissions per capita CO_2 are measured in metric tons per capita and included since nuclear power is considered by some as a low carbon generation source although characterized by high lifecycle emissions [40].⁸

After explaining the construction of our polytomous dependent variable and describing our utilized categorical measure for the level of democratic development, Table 1 shows how the three different nuclear energy statuses descriptively relate to the three previously defined levels of democratic quality. Our analysis covers 166 countries and 58 years which results in 9,628 country-years. However, due to the unbalanced nature of our panel time series data set, a total number of 8,266 observations is available for the combination of the level of democratic development and the nuclear energy status. The row frequencies

⁸Evaluating 103 lifecycle studies of greenhouse gas-equivalent emissions for nuclear power plants, Sovacool [40] identifies the range of emissions for nuclear energy over the lifetime of a plant from 1.4 kg CO2-eq/MWh to 288 kg CO2-eq/MWh, with an average estimate of 66 kg CO2-eq/MWh. Lenzen [41] identifies the greenhouse gas intensities for light and heavy water reactors from 10 and 130 kg CO2-eq/MWh, with an average of 65 kg CO2-eq/MWh. The variability is due to different technologies and methodological differences between process chain analysis (PCA) and input-output analysis (IOA), the two main approaches used to assess emissions in a lifecycle analysis [42]. For an assessment of nuclear power regarding various sustainability development criteria, see Verbruggen et al. [43].

		Polity IV Project					
	F	PF	NF	Total			
Non-nuclear	2,833	1,047	2,966	6,846			
	(41.38)	(15.29)	(43.32)	(100)			
Construction	50	28	96	174			
	(28.74)	(16.09)	(55.17)	(100)			
Nuclear	1,111	26	109	1,246			
	(89.17)	(2.09)	(8.75)	(100)			
Total	3,994	1,101	3,171	8,266			
	(48.32)	(13.32)	(38.36)	(100)			

Table 1: Frequency table for the nuclear energy statuses at each level of democratic quality

Notes: F, PF, and NF correspond to democratically free, partly free, and not free for the measure of democratic quality D1 from Polity IV Project, respectively. Row percentages are in parentheses.

thus represent how many observations for the respective level of democratic development fall into the respective category of nuclear energy status. Non-nuclear statuses dominate the sample. Within the given period, the majority of the total observations in our sample has been evaluated as democratically free. Within the non-nuclear group, more than 40% of the observations are either democratically free or not free. Democratically less free countries however clearly dominate the construction periods whereas "nuclear" statuses are characterized by higher percentages of democratically free countries. The nuclear statuses which follow the initial introduction of nuclear power not only occur more frequent in our sample but also correspond chronologically to the subsequent periods. It is thus not surprising that nuclear statuses are characterized by higher percentages of democratically free countries against the background of the tendency of overall increasing democracy levels of societies in the last several years and the transition to democracy of countries during the period 1985 and 1995 [44, 45].

Table 2 reports the mean values and standard deviations for the explanatory variables in their respective nuclear status. The continuous democracy variable D1 is the normalized democracy measure from Polity2 index running from 0 to 100 with greater values representing higher levels of democratic development [33]. GDP is GDP per capita and measured in constant 2010 USD. Urbanization U is measured as the share of the population living in urban areas. Electric power consumption E is measured in kWh per

	D1	GDP	U	E	EI	FFR	CO_2
Non-nuclear	49.76	8,670.07	45.35	2,330.84	-106.16	3.67	3.75
	(36.04)	(17, 834.67)	(24.51)	(3,731.93)	(642.98)	(10.33)	(7.96)
Construction	42.24	8,329.60	54.62	2,372.46	-29.06	4.86	5.48
	(34.71)	(8,670.71)	(16.63)	(2,003.72)	(166.84)	(9.12)	(3.90)
Nuclear	87.19	20,902.98	67.61	5,451.87	37.52	1.15	7.84
	(24.02)	(16, 974.32)	(16.91)	(4,056.77)	(37.68)	(2.66)	(4.76)

 Table 2: Arithmetic means and standard deviations of the seven explanatory variables for the three nuclear energy statuses

Notes: Arithmetic mean is shown and the standard deviation is in parentheses. Data on D1 are taken from the Polity IV Project. Data on GDP, U, E, EI, FFR, and CO_2 are taken from the World Development Indicators (WDI) database from The World Bank (last updated 24 April 2019).

capita. Energy imports EI is measured in % of primary energy use. Fossil fuel rents FFR are the sum of oil rents, natural gas rents, coal rents and measured in percentage of GDP. CO₂ emissions per capita CO_2 are measured in metric tons per capita.

Figures 1 to 4 show the boxplots for all variables in three nuclear energy statuses. The normalized democracy measure from the Polity2 index D1 is on average twice as

Figure 1: Boxplots for D1 and GDP for the three nuclear energy statuses



high for the nuclear statuses compared to the construction statuses. D1 is 1.75 times higher for the nuclear statuses compared to the non-nuclear statuses. D1 is 1.18 times higher for the non-nuclear statuses compared to the construction statuses. Average GDP per capita is the largest for the nuclear statuses and the smallest for the construction statuses. Similarly, the range of GDP per capita is the largest for the nuclear statuses. Urbanization levels are highest for the nuclear and construction statuses compared to





the non-nuclear statuses. Urbanization levels vary the most for the non-nuclear statuses compared to the nuclear and construction statuses. Average electric power consumption per capita is the highest for the nuclear statuses; they also show the most variability for electric power consumption. For the non-nuclear statuses, electric power consumption per capita is on average almost identical compared to the construction statuses. Nuclear

Figure 3: Boxplots for EI and FFR for the three different nuclear energy statuses



statuses are characterized by positive values for the net energy imports indicator, whereas both non-nuclear and construction statuses are associated with negative values. Nuclear statuses on average tend to import energy, whereas non-nuclear and construction statuses are characterized by net energy exports. Construction statuses are characterized by the highest value for fossil fuel rents and also show the greatest variability for fossil fuel rents. Non-nuclear statuses have higher fossil fuel rents on average compared to nuclear statuses. Average CO_2 emissions per capita for the nuclear statuses are 1.43 (2.09) times higher

Figure 4: Boxplots for CO_2 for the three nuclear energy statuses



compared to the construction (non-nuclear) statuses, whereas the average CO_2 emissions per capita for the construction statuses are 1.46 times higher compared to the non-nuclear statuses.

3.2 Methodological Approach

Our categorical response variable contains more than two outcomes. The multinomial logistic regression approach thus is used to model relationships between a polytomous outcome variable and a set of predictor variables. The multinomial logit model builds on the binary logit model, but the factors which affect the outcomes are determined simultaneously which increases the efficiency of the estimates. The multinomial logistic regression approach uses the maximum likelihood estimation technique to establish the probability of group membership. The categories of the outcome variable are restricted to be unordered and based on the assumption of the independence of irrelevant alternatives (IIA) stating that the inclusion or exclusion of categories does not affect the relative risks associated with the remaining categories [46]. Hence, we utilize a multinomial logistic regression approach to analyze if a countries' choice to "go nuclear" is significantly influenced by the level of democratic development. In the multinomial logit model, the log-odds ratio that country i will fall in response category j relative to the reference category J is assumed to follow a linear model:

$$\eta_{ij} = \log\left(\frac{\pi_i^{(j)}}{\pi_i^{(J)}}\right) = \alpha^{(j)} + \beta_1^{(j)} X_{1i} + \dots + \beta_k^{(j)} X_{ki},\tag{1}$$

where π_i is the probability for outcome j in the i = 1, ..., n countries, $\alpha^{(j)}$ is a constant, $\beta_1^{(j)}, ..., \beta_k^{(j)}$ are the k regression coefficients, for the j = 1, ..., J - 1 outcomes, and $X_{1i}, ..., X_{ki}$ are the k explanatory variables.

We model a countries' nuclear energy strategy in which economies face the following j choices in the defined categorical dependent variable N: not using nuclear (j = 0,"non-nuclear"), constructing a nuclear power plant (j = 1,"construction"), and having at least one nuclear power plant fully operational (j = 2,"nuclear"). Since we specify "non-nuclear" as our reference category, we obtain a model for the log-odds of choosing "construction" over "non-nuclear":

$$\eta_{i1} = \log\left(\frac{\pi_i^{(1)}}{\pi_i^{(0)}}\right) = \alpha^{(1)} + \beta_1^{(1)}F_i + \beta_2^{(1)}PF_i + \beta_3^{(1)}NF_i + \beta_4^{(1)}W_i + \beta_5^{(1)}GDP_i + \beta_6^{(1)}U_i + \beta_7^{(1)}E_i + \beta_8^{(1)}EI_i + \beta_9^{(1)}FFR_i + \beta_{10}^{(1)}CO_{2i},$$
(2)

and a second model for the log-odds of choosing "nuclear" over "non-nuclear":

$$\eta_{i2} = \log\left(\frac{\pi_i^{(2)}}{\pi_i^{(0)}}\right) = \alpha^{(2)} + \beta_1^{(2)}F_i + \beta_2^{(2)}PF_i + \beta_3^{(2)}NF_i + \beta_4^{(2)}W_i + \beta_5^{(2)}GDP_i + \beta_6^{(2)}U_i + \beta_7^{(2)}E_i + \beta_8^{(2)}EI_i + \beta_9^{(2)}FFR_i + \beta_{10}^{(2)}CO_{2i},$$
(3)

where the factor variables F_i , PF_i , and NF_i correspond to the transformed normalized Polity2 index D1. We include a dummy variable indicator for nuclear warheads W, GDP per capita GDP, the share of urban population U, electric power consumption (kWh per capita) E, net energy imports (% of energy use) EI, fossil fuel rents (% of GDP) FFR, and CO₂ emissions (metric tons per capita) CO_2 .

	List of used	acronyms	
IIA	Independence of Irrelevant Alternatives	PRIS	Power Reactor Information System
GLMM	Generalized Linear Mixed Model	\mathbf{PSI}	Political Stability Index
IAEA	International Atomic Energy Agency	PITF	Political Instability Task Force
LR	Likelihood Ratio	RRR	Relative Risk Ratio
MLP	Multi-Level Perspective	WNA	World Nuclear Association
NPT	Nuclear Non-Proliferation Treaty	WDI	World Development Indicators
	List of used	variables	
$\overline{D_1}$	Continuous democracy indicator	U	Urbanization,
F	Democratically free	E	Electric power consumption per capita
PF	Democratically partly free	EI	Energy imports
NF	Democratically not free	FFR	Fossil fuel rents
W	Nuclear warhead possession	CO_2	CO_2 emissions per capita
GDP	GDP per capita		

 Table 3:
 Nomenclature

4 Empirical Results

We start conducting model specification tests and test for the assumption of independence of irrelevant alternatives (IIA) of the multinomial logit model [47].⁹ We first test if combining our dependent categories would increase the efficiency of our estimates. We thus test if a pair of outcomes is indistinguishable using both a Likelihood Ratio (LR) and Wald test. For both the LR and Wald test and for every pair of outcomes we can reject the null hypothesis that alternatives can be collapsed. This indicates that our models are efficiently defined in terms of the dependent categories. To assess if the effect of an independent variable equals zero across all equations we use both a LR and Wald test, to test the null hypothesis that all of the coefficients associated with an independent variable are simultaneously equal to zero across all equations. Again, for both the LR and Wald test, we can reject the null hypothesis that all coefficients for each regressor are equal to zero. Third, we test if the inclusion or exclusion of categories does not affect the relative risks associated with the remaining categories to evaluate if the outcome categories for

⁹The results of all tests are available upon request. We inspect the correlations among variables to evaluate if multicollinearity affects our analyses. The results are in Table 1 in the appendix and indicate overall relatively low correlation among the variables. However, the correlation between GDP and U, GDP and E, GDP and CO_2 , U and E, U and CO_2 , and E and CO_2 exceeds 0.5.

the model have the property of IIA with the Hausman test of IIA [46]. The test statistics for all three categories are negative which suggests that the IIA has not been violated.¹⁰

We interpret the estimated parameters in Table 4 from the multinomial logistic regression approach relative to the reference group "non-nuclear". The results show the exponentiated estimates for the log-odds ratios associated with equations (2) and (3), respectively, which are interpreted in terms of the relative risk ratios (RRR). The RRR indicate how the probability of choosing alternative j relative to the reference group changes if the corresponding variable increases by one unit, ceteris paribus. We thus interpret the respective category of interest (partly free or not free) relative to the base category (free). In addition to our baseline estimations, we test if the results are driven by countries such as the United States or Russia which have nuclear weapons and nuclear power prior to the start of our study period.

	(1) Baseline		(2) Baseline	no USA	(3) Baseline n	(3) Baseline no Russia		
	$\overline{\begin{array}{c} \text{Construction} \\ (j=1) \end{array}}$	Nuclear $(j=2)$	$\overline{\begin{array}{c} \text{Construction} \\ (j=1) \end{array}}$	Nuclear $(j=2)$	$\overline{\begin{array}{c} \text{Construction} \\ (j=1) \end{array}}$	Nuclear $(j=2)$		
PF	0.842	0.117^{a}	0.841	0.118^{a}	0.865	0.0859^{a}		
$\frac{NF}{W}$	$\frac{2.526^{a}}{4.424^{a}}$	$\frac{0.275^{a}}{20.56^{a}}$	$\frac{2.523^{a}}{4.333^{a}}$	0.280^a 18.22^a	$\frac{2.521^{a}}{4.452^{a}}$	0.291^a 16.68^a		
GDP	1.000	1.000^{a}	1.000	1.000^{a}	1.000	1.000^{a}		
E	1.028^{-1} 1.000^{b}	1.025^{a} 1.000^{a}	1.028^{-1} 1.000^{b}	1.025^{a} 1.000^{a}	1.029^{-1} 1.000^{c}	1.022^{a} 1.000^{a}		
EI FFR	1.002^{a} 1.029 ^a	1.008^{a}	1.002^{a} 1.029 ^a	1.008^{a}	1.002^{a} 1.029 ^a	1.008^{a}		
CO_2	1.045	0.963^{a}	1.045	0.959^{a}	1.045	0.962^a		
cons	0.00450^{a}	0.0544^{a}	0.00451^{a}	0.0547^{a}	0.00445^{a}	0.0571^{a}		
$\frac{\rm N}{\rm Pseudo}\; \rm R^2$	$4844 \\ 0.279$		4789 0.259)	4821 0.274			

Table 4: Estimations (1) to (3) with categorized democratic quality from Polity IV Project

Notes: RRR is shown. Superscripts a, b, and c represent significance at 1%, 5%, and 10%, respectively. Democratically free F is the base category. The reference group is non-nuclear (j = 0).

¹⁰The Hausman test of IIA in general unfortunately provides rather inconsistent results thus providing little guidance whether the IIA assumption is violated or not. Based on simulations, Cheng and Long [48] show that the size properties of commonly used IIA tests depend on the data structure for the predictor variables. As a results, it is not uncommon that IIA tests often reject the assumption when the alternatives seem distinct and often fail to reject IIA when the alternatives can reasonably be viewed as close substitutes even in well-specified models [48]. They conclude that "[...] tests of the IIA assumption that are based on the estimation of a restricted choice set are unsatisfactory for applied work."

All estimated parameters are statistically significant at least at the 10% level, except for the parameters on PF, GDP, and CO_2 in estimation (1), (2), and (3) for j = 1. The parameters on FFR are only statistically significant for j = 1 in estimation (1), (2), and (3). We begin interpreting the baseline estimation and the RRR for j = 1. The RRR for NF is above unity. Thus, for democratically not free countries compared to democratically free countries the relative risk of being in the construction group relative to the non-nuclear group would be expected to increase, ceteris paribus. Given all other variables held constant and compared to democratically free countries, the probability that democratically not free countries being in the construction group instead of in the non-nuclear group increases by 153% ($(2.526-1) \times 100$). On the contrary, the RRR for the statistically not significant parameter on PF is below unity which indicates that for democratically partly free countries compared to democratically free countries the relative risk of being in the construction group relative to the non-nuclear group would be expected to decrease, ceteris paribus. On the contrary, the RRR for j = 2 for both PF and NF are below unity. Thus, for both democratically partly free and not free countries compared to democratically free countries the relative risk of being in the nuclear group relative to the non-nuclear group would be expected to decrease, ceteris paribus. Given all other variables held constant and compared to democratically free countries, the probability that democratically partly free countries are in the nuclear group instead of in the non-nuclear group decreases by 88% $((1-0.117) \times 100)$. Similarly, given all other variables held constant and compared to democratically free countries, the probability of democratically not free countries being in the nuclear group instead of in the non-nuclear group decreases by 73% ((1 - 0.275) × 100).

The RRR for W are above unity in both j = 1 and j = 2 yet differing substantially in magnitude. Thus for countries which possess at least one nuclear warhead compared to countries without a nuclear warhead, the relative risk of being in the construction group and nuclear group, respectively, relative to the non-nuclear group would be expected to increase, ceteris paribus. Given all other variables held constant and compared to countries without a nuclear warhead, the probability that countries which possess at least one nuclear warhead are in the construction group instead of in the non-nuclear increases by 342% ($(4.424 - 1) \times 100$). Similarly, given all other variables held constant and compared to countries without a nuclear warhead, the probability that countries which possess at least one nuclear warhead are in the nuclear group instead of in the non-nuclear increases by 1956% ($(20.56 - 1) \times 100$).

The RRR for U and EI are above unity for j = 1 as well as for j = 2. Thus, a one unit increase in U or EI increases the probability that to "go nuclear" (to be nuclear) is chosen instead of non-nuclear (non-nuclear). Countries which become rather energy importers are more likely to both construct and continue to use nuclear power. The RRR for FFR is above unity for j = 1. Thus, a one unit increase in FFR increases the probability that to "go nuclear" is chosen instead of non-nuclear. Increases in fossil fuel rents, which effectively function as an asset like any other stock of capital, increase the probability of entering construction. The RRR for CO_2 is below unity for j = 2. Thus, a one unit increase in CO_2 decreases the probability that to be nuclear is chosen instead of non-nuclear. The result indicates that choosing to continue using nuclear power historically is not mainly motivated by CO_2 emission reduction efforts to tackle climate change. The abatement of greenhouse gas emissions moreover has been identified only recently as a potential factor emerging in the environmental policy realm (Sovacool and Valentine, 2010). Considering the national development and both energy transitions and environmental indicators only, the impact of a one unit increase on the probability that to "go nuclear" and to be nuclear is chosen instead of non-nuclear is the greatest in magnitude for U. If two countries are identical except for their urbanization levels, the country with higher urbanization is more likely to choose to "go nuclear" and to be nuclear than the country with lower urbanization. The RRR for both GDP and E are equal to one for j = 1 as well as for j = 2 which indicates a rather unsubstantial effect of both variables for defining the nuclear energy strategy chosen over time. Considering the magnitude of all of the estimated parameters, both democracy and possession of a nuclear warhead tend to have the largest impact on the nuclear energy strategy chosen over time. Overall, the results do not vary substantially when we exclude the United States or Russia.

Figure 5 shows the predicted probabilities from our baseline estimation for democratic quality while holding all other variables at their means. The left plot illustrates the predicted probabilities for the introduction of nuclear power at each level of democratic freedom - free F, partly free PF, not free NF - while holding all other variables at their means.





Notes: Left plot indicates categorized levels and right plot indicates continuous levels based on the Polity IV Project democracy measure. F, PF, and NF are democratically free, partly free, and not free, respectively. Continuous levels of democratic quality on the horizontal axis in the right plot correspond to D1. Higher values represent higher levels of democratic development [33].

The predicted probability for introducing nuclear power increases with decreasing levels of democratic freedom. While the predicted probability for the introduction of nuclear power is slightly higher for democratically free countries than for democratically partly free countries considering the point estimates, the upper bound of the confidence interval for democratically partly free countries is considerably greater in magnitude compared to democratically free countries. The right plot indicates the predicted probabilities for continuous levels of democratic quality while holding all other variables at their means. Greater values represent higher levels of democratic development. The predicted probability for introducing nuclear power increases with decreasing levels of democratic freedom. Generally, democracy effects and the possession of a nuclear warhead tend to dominate a countries' nuclear energy strategy chosen over time instead of national development and both energy transitions and environmental indicators with the exception of urbanization levels.

5 Robustness

We conduct several robustness checks. First, we create a dummy for the transitional years covering the period 1989 to 1992 to capture year effects which affect all countries between the fall of the Berlin Wall and the collapse of the Soviet Union. We also use a specification with year effects to capture the influence of aggregate time-series trends. Second, we gradually increase a parsimonious specification which only includes the democracy control factor variable with the other relevant predictors until we arrive at the baseline specification given in equations (2) and (3), respectively. Third, we redo the entire analysis but use the democracy measure from the normalized Polity2 index in continuous modes. Table 5 presents the results with an additional dummy covering the period 1989 to 1992 as well as with year effects included.

 Table 5: Estimations (4) and (5) with categorized democratic quality from Polity IV Project with transitional period dummy and year effects

	(4) Dum	imy	(5) Year E	(5) Year Effects			
	$\overline{\text{Construction}} $ $(j=1)$	Nuclear $(j=2)$	$\overline{\begin{array}{c} \text{Construction} \\ (j=1) \end{array}}$	Nuclear $(j=2)$			
PF	0.843	0.117^{a}	1.076	0.113^{a}			
NF	2.526^{a}	0.275^{a}	1.979^{b}	0.253^{a}			
W	4.438^{a}	20.59^{a}	4.299^{a}	21.59^{a}			
GDP	1.000	1.000^{a}	1.000^{b}	1.000^{a}			
U	1.029^{a}	1.023^{a}	1.031^{a}	1.024^{a}			
E	1.000^{b}	1.000^{a}	1.000	1.000^{a}			
EI	1.002^{a}	1.008^{a}	1.003^{a}	1.008^{a}			
FFR	1.030^{a}	0.983	1.073^{a}	0.995			
CO_2	1.045	0.963^{a}	0.979	0.952^{a}			
Dummy 1989 - 1992	1.148	1.051	-	-			
Year Effects	no	no	yes	yes			
cons	0.00443^{a}	0.0541^{a}	0.0950^{a}	0.0318^{a}			
N	4844		4844				
Pseudo \mathbb{R}^2	0.279		0.302				

Notes: RRR is shown. Superscripts a, b, and c represent significance at 1%, 5%, and 10%, respectively. Democratically free F is the base category. The reference group is non-nuclear (j = 0).

The results are identical to our baseline estimation in significance and magnitude for all of the estimated parameters in both j = 1 and j = 2. The time dummy for 1989 to 1992 is statistically not significant for either j = 1 or j = 2, which indicates that the effects of the transitional period into the post-cold war era do not significantly impact the nuclear energy strategy chosen over time. If we capture the influence of aggregate time-series trends with year effects, significance and magnitude do not change. We can reject the null hypothesis that the coefficients for all years are jointly equal to zero at the 10% level significance level. The main implications remain the same. For democratically not free countries compared to democratically free countries the relative risk of being in the construction group relative to the non-nuclear group would be expected to increase, ceteris paribus. Similarly, countries which possess at least one nuclear warhead compared to countries without a nuclear warhead, are more likely to be in the construction group and nuclear group, respectively, relative to the non-nuclear group.

Table 6 shows the results of our second robustness check which do not alter the main implications. When we gradually increase a parsimonious specification which only **Table 6:** Estimations (6) to (12) with categorized democratic quality from Polity IV Project

	(6)		(7)		(8)		(9)		
	Construction (j=1)	n Nuclear $(j=2)$	$\begin{array}{c} \text{Constructio} \\ (j=1) \end{array}$	n Nuclear $(j=2)$	$\begin{array}{c} \text{Construction} \\ (j=1) \end{array}$	n Nuclear $(j=2)$	$\begin{array}{c} \text{Construction} \\ (j=1) \end{array}$	n Nuclear $(j=2)$	
PF	1.515^{c}	0.0633^{a}	1.539^{c}	0.0719^{a}	1.359	0.101^{a}	1.502	0.113^{a}	
NF	1.834^{a}	0.0937^{a}	1.833^{a}	0.0933^{a}	2.102^{a}	0.126^{a}	2.557^{a}	0.172^{a}	
W			3.372^{a}	23.42^{a}	3.660^{a}	21.80^{a}	4.187^{a}	22.48^{a}	
GDP					1.000	1.000^{a}	1.000^{a}	1.000	
U							1.031^{a}	1.033^{a}	
cons	0.0176^{a}	0.392^{a}	0.0172^{a}	0.312^{a}	0.0142^{a}	0.228^{a}	0.00340^{a}	0.0400^{a}	
N	8266 8266		6	705	5	7045			
Pseudo \mathbb{R}^2	0.128		0.192		0.203		0.238		
	(10)		(11)		(12)		(1)		
	Construction	n Nuclear (Constructio	n Nuclear	Construction	n Nuclear	Construction	n Nuclear	
	(j=1)	(j=2)	(j=1)	(j=2)	(j=1)	(j=2)	(j=1)	(j=2)	
PF	0.942	0.110^{a}	0.994	0.120^{a}	0.874	0.122^{a}	0.842	0.117^{a}	
NF	2.603^{a}	0.154^{a}	3.008^{a}	0.258^{a}	2.615^{a}	0.269^{a}	2.526^{a}	0.275^{a}	
W	4.583^{a}	20.80^{a}	4.370^{a}	19.01^{a}	4.512^{a}	19.44^{a}	4.424^{a}	20.56^{a}	
GDP	1.000	1.000^{a}	1.000	1.000^{a}	1.000	1.000^{a}	1.000	1.000^{a}	
U	1.030^{a}	1.025^{a}	1.030^{a}	1.021^{a}	1.029^{a}	1.021^{a}	1.028^{a}	1.023^{a}	
E	1.000	1.000^{a}	1.000	1.000^{a}	1.000	1.000^{a}	1.000^{b}	1.000^{a}	
EI			1.001	1.008^{a}	1.002^{a}	1.007^{a}	1.002^{a}	1.008^{a}	
FFR					1.032^{a}	0.973^{c}	1.029^{a}	0.983	
CO_2							1.045	0.963^{a}	
cons	0.00431^{a}	0.0663^{a}	0.00410^{a}	0.0565^{a}	0.00430^{a}	0.0587^{a}	0.00450^{a}	0.0544^{a}	
N	486	6	486	0	4860		4821		
Pseudo \mathbb{R}^2	0.23	4	0.27	4	0.27	0.276		0.274	

Notes: RRR is shown. Superscripts a, b, and c represent significance at 1%, 5%, and 10%, respectively. Democratically free F is the base category. The reference group is non-nuclear (j = 0).

includes the democracy control factor variable with the other relevant predictors, the results still suggest that i) in particular for democratically not free countries compared to democratically free countries the relative risk of being in the construction group relative to the non-nuclear group would be expected to increase and ii) that for both democratically partly free and not free countries compared to democratically free countries the relative risk of being in the nuclear group relative to the non-nuclear group would be expected to democratically free countries the relative risk of being in the nuclear group relative to the non-nuclear group would be expected to decrease. Moreover, similar to our baseline estimation, the RRR for W for both j = 1 and j = 2 is always above unity which indicates that countries which possess at least one nuclear warhead compared to countries without, the relative risk of being in the construction and nuclear group, respectively, relative to the non-nuclear group would be expected to increase.

The results for our last robustness check in which we redo the entire analysis utilizing democratic quality in continuous modes, are presented in Tables 7 to 9. If we redo the entire analysis but use the democracy measures from the Polity IV Project in continuous modes, the results indicate that if a country increases its democracy level, we would expect this country to be more likely to choose non-nuclear over going nuclear which supports our main findings.

	(13) Baseline		(14) Baseline	no USA	(15)Baseline no Russia		
	Construction $(j=1)$	Nuclear $(j=2)$	$\begin{array}{c} \text{Construction} \\ (j=1) \end{array}$	Nuclear $(j=2)$	Construction $(j=1)$	Nuclear $(j=2)$	
D1	0.991^{a}	1.024^{a}	0.991^{a}	1.023^{a}	0.991^{a}	1.023^{a}	
W	4.614^{a}	21.89^{a}	4.525^{a}	19.57^{a}	4.756^{a}	19.07^{a}	
GDP	1.000	1.000^{a}	1.000	1.000^{a}	1.000	1.000^{a}	
U	1.027^{a}	1.021^{a}	1.027^{a}	1.021^{a}	1.027^{a}	1.021^{a}	
E	1.000^{c}	1.000^{a}	1.000^{c}	1.000^{a}	1.000^{c}	1.000^{a}	
EI	1.002^{a}	1.007^{a}	1.002^{a}	1.008^{a}	1.002^{a}	1.007^{a}	
FFR	1.030^{a}	0.995	1.030^{a}	0.997	1.030^{a}	0.987	
CO_2	1.044	0.962^{a}	1.045	0.959^{a}	1.044	0.962^{a}	
cons	0.0118^{a}	0.00727^{a}	0.0118^{a}	0.00748^{a}	0.0116^{a}	0.00780^{a}	
N	4844		4789	4789		4821	
Pseudo \mathbb{R}^2	0.279		0.259)	0.273		

Table 7: Estimations (13) to (15) with continuous democratic quality from Polity IV Project

Notes: RRR is shown. Superscripts a, b, and c represent significance at 1%, 5%, and 10%, respectively. D1 is the continuous democracy measure. The reference group is non-nuclear (j = 0).

	(16) Dur	nmv	(17) Year Effects			
	Construction	Nuclear	Construction	Nuclear		
	(j=1)	(j=2)	(j=1)	(j=2)		
<i>D</i> 1	0.991^{a}	1.024^{a}	0.995^{c}	1.024^{a}		
W	4.627^{a}	21.92^{a}	4.490^{a}	22.99^{a}		
GDP	1.000	1.000^{a}	1.000^{b}	1.000^{a}		
U	1.027^{a}	1.021^{a}	1.030^{a}	1.022^{a}		
E	1.000^{c}	1.000^{a}	1.000	1.000^{a}		
EI	1.002^{a}	1.008^{a}	1.003^{a}	1.008^{a}		
FFR	1.030^{a}	0.995	1.076^{a}	1.006		
CO_2	1.044	0.962^{a}	0.977	0.952^{a}		
Dummy 1989 - 1992	1.160	1.060	-	-		
Year Effects	no	no	yes	yes		
cons	0.0116^{a}	0.00722^{a}	0.179^{a}	0.00375^{a}		
Ν	4844	:	4844			
Pseudo \mathbb{R}^2	0.280)	0.304			

Table 8: Estimations (16) and (17) with continuous democratic quality from Polity IV Projectwith transitional period dummy and year effects

Notes: RRR is shown. Superscripts a, b, and c represent significance at 1%, 5%, and 10%, respectively. D1 is the continuous democracy measure. The reference group is non-nuclear (j = 0).

Based on our empirical analysis, the overall results indicate the following: We robustly find i) that in particular for democratically not free countries compared to democratically free countries the probability of being in the construction group relative to the non-nuclear group would be expected to increase whereas this probability decreases for the nuclear group and robustly for all estimations ii) that countries which possess at least one nuclear warhead compared to countries without are more likely to choose to construct a nuclear power plant and to use nuclear power, respectively, instead of not using nuclear power at all. Overall, the estimated probability for being a democratically not free country in the construction group instead of in the non-nuclear group ranges from 83% to 200%. The estimated probabilities for construction and the continued use of nuclear for countries possessing at least one nuclear warhead compared to countries that do not have a nuclear warhead ranges from 228% to 384% in the case of construction and from 1568% to 2635% in the case of the continued use of nuclear. Again, possession of a nuclear warhead tends to encourage the continued use of nuclear energy.

Our results regarding the democratic realm are broadly supported by Yamamura [31] for freedom of expression and free media which significantly influences views on the

	(18)		(19)	(19)			(21)		
	Construction (j=1)	Nuclear $(j=2)$	$\begin{array}{c} \text{Construction} \\ (j=1) \end{array}$	Nuclear $(j=2)$	$\begin{array}{c} \text{Construction} \\ (j=1) \end{array}$	Nuclear $(j=2)$	$\begin{array}{c} \text{Construction} \\ (j=1) \end{array}$	Nuclear $(j=2)$	
D1	0.994^{a}	1.041^{a}	0.994^{a}	1.041^{a}	0.993^{a}	1.036^{a}	0.990^{a}	1.030^{a}	
W			3.281^{a}	27.35^{a}	3.688^{a}	25.44^{a}	4.334^{a}	24.68^{a}	
GDP					1.000	1.000^{a}	1.000^{a}	1.000	
U							1.030^{a}	1.031^{a}	
cons	0.0332^{a}	0.0103^{a}	0.0323^{a}	0.00811^{a}	0.0301^{a}	0.0101^{a}	0.00935^{a}	0.00317^{a}	
N	8266	8266 8266		7055		7045			
Pseudo \mathbb{R}^2	R^2 0.151		0.210	0.216		0.218		0.247	
	(22)		(23)	(23)		(24)		(13)	
	Construction	Nuclear	Construction	Nuclear	Construction	Nuclear	Construction	Nuclear	
	(j=1)	(j=2)	(j=1)	(j=2)	(j=1)	(j=2)	(j=1)	(j=2)	
D1	0.990^{a}	1.029^{a}	0.988^{a}	1.024^{a}	0.990^{a}	1.024^{a}	0.991^{a}	1.024^{a}	
W	4.835^{a}	22.30^{a}	4.601^{a}	20.50^{a}	4.702^{a}	20.78^{a}	4.614^{a}	21.89^{a}	
GDP	1.000	1.000^{a}	1.000	1.000^{a}	1.000	1.000^{a}	1.000	1.000^{a}	
U	1.029^{a}	1.024^{a}	1.029^{a}	1.019^{a}	1.027^{a}	1.019^{a}	1.027^{a}	1.021^{a}	
E	1.000	1.000^{a}	1.000	1.000^{a}	1.000	1.000^{a}	1.000^{c}	1.000^{a}	
EI			1.001	1.007^{a}	1.002^{a}	1.007^{a}	1.002^{a}	1.007^{a}	
FFR					1.032^{a}	0.984	1.030^{a}	0.995	
CO_2							1.044	0.962^{a}	
cons	0.0118^{a}	0.00506^{a}	0.0132^{a}	0.00737^{a}	0.0117^{a}	0.00777^{a}	0.0118^{a}	0.00727^{a}	
N	4866	;	4860)	4860	4860		4844	
Pseudo \mathbb{R}^2	0.242	1	0.275	5	0.277	7	0.279	0.279	

Table 9: Estimations (18) to (24) with continuous democratic quality from Polity IV Project

Notes: RRR is shown. Superscripts a, b, and c represent significance at 1%, 5%, and 10%, respectively. D1 is the continuous democracy measure. The reference group is non-nuclear (j = 0).

security of nuclear power plants: Citizens tend to disagree that nuclear power plants are properly secured against accidents when the political setting assures both freedom of expression and media to a greater extent. In terms of nuclear weapons, our results contrast Fuhrmann [13] who finds the impact of nuclear weapons exploration on the likelihood for construction to be statistically not significant.

6 Conclusion

This paper analyzes how a countries' choice to introduce nuclear power is influenced by the level of democratic development. Our empirical analysis is based on a panel time series data set with 166 countries covering the period 1960 to 2017. We utilize a multinomial logistic regression approach to evaluate how different stages of a countries' nuclear energy strategy - not using nuclear power at all, construction, and continued use of nuclear power - relates to different levels of democratic development. Our empirical results robustly show that historically, democratically not free countries compared to democratically free countries are more likely to introduce nuclear power instead of not using nuclear power at all. Moreover, countries possessing at least one nuclear warhead compared to countries without a nuclear warhead are more likely to decide to use nuclear power. Democracy effects and possession of a nuclear warhead have a significant effect on the nuclear energy strategy chosen over time.

The introduction of nuclear energy thus tends to be more likely under conditions where political and public debate are minimized which more easily enables the implementation of governmental programs which might run contrary to public interest. Decisions regarding nuclear power are moreover influenced by private and/or governmental technocracy which can overpower democratic steering and control processes detrimentally. When technocracy can influence its regulators, it can also impact deliberative forums and public engagement to foster incumbent nuclear policy which is in contrast to the encouragement of sustainable development policy. Government policy however can address these issues by a more pronounced public involvement in decisions regarding nuclear power and by increasing stakeholder involvement at all aspects of the nuclear fuel cycle including uranium mining, radioactive waste management, location of new nuclear power plants, emergency situations and rehabilitation of contaminated territories [43].

The political setting moreover tends to dominate and overpower both the social and economic dimension at least in less democratic environments when it comes to nuclear energy deployment. Certainly, nuclear power requires a specific type of governance due to the very specific safety requirements of nuclear power plants and their impact on society. The very specific conditions in large-scale energy infrastructure projects such as institutional exceptions then tend to have an impact on the practices and institutions which define the governance of a project [49]. We thus provide empirical evidence on how certain political environments favor the implementation of large-scale energy infrastructures with lifespans over decades. It becomes more difficult for countries to move towards decentralized energy technologies that are considered to more readily organize and enable distributed political and economic power which effectively can create a technological lockin, if the nuclear electricity industry moreover is both highly concentrated and connected with related policy decisions [50]. How to overcome a potential nuclear lock-in partly induced due to certain political environments is a conversely related question which also derives from our empirical analysis. Aspects of democratic quality similarly impact the nuclear energy strategy for either continuity or disruption of nuclear power. In particular against the background of the democratic deficits associated with nuclear power, qualities of democracy appear to be an inconspicuous yet highly relevant factor connected also to nuclear discontinuity. Finland can be seen as an exception as nuclear new builds are planned in this country [20].

Due to the dual-use dilemma nuclear energy faces, our analyses moreover have implications beyond energy and environmental policy addressing international relations, conflict, and security issues. Nuclear weapon aspirations or possession thereof might be accompanied by the pursuit of nuclear power, and vice versa. The ownership of nuclear weapons then can eventually impede a nuclear phase out globally. But similarly, the desire for nuclear warheads can motivate countries to construct nuclear power plants. In countries such as China, India, and Pakistan, the introduction of nuclear power for civil purposes was only possible in conjunction with developing nuclear weapons through mobilizing extraordinary political will and resources [29]. The synergies between military use of nuclear power and electricity generation in countries such as Iran could result in a multi-nuclear Middle East with both Saudi Arabia and Egypt most likely being candidates choosing to "go nuclear" very soon in a response to a potential Iranian warhead [51].

We suggest that future research should emphasize additional geopolitical and military aspects of nuclear power deployment. Specific aspects of qualities of democracy and their impacts on different nuclear trajectories also need to be analyzed. For example, are countries within proximity of hostile countries with nuclear intercontinental ballistic missiles more likely to construct nuclear power plants? Does country membership in a defense alliance affect its nuclear power deployment? Connecting the different dimensions of democratic quality with the utilization of nuclear energy is expected to provide further insights into the current and future development of nuclear power. Building upon the work from Johnstone and Stirling [20], it would be of particular interest to investigate how different aspects of democratic quality impact the decision to phase out nuclear power. In this regard, we suggest considering the potential effects of specific democratic elements such as the rule of law, participation, competition, both vertical and horizontal accountability, respect for civil and political freedoms, progressive implementation of greater political equality, and responsiveness on which democracies diverge in terms of quality to evaluate which democratic characteristics are the most important to explain the difference in nuclear energy trajectories among countries.

References

- F. Lévêque, The Economics and Uncertainties of Nuclear Power, Cambridge University Press, Cambridge, United Kingdom, 2014, https://doi.org/10.1017/ CB09781316095782.
- [2] C. von Hirschhausen, Nuclear Power in the 21st Century An Assessment (Part I), DIW Discussion Papers, No. 1700. DIW Berlin, Berlin, 2017, Retrieved from https://www.diw.de/documents/publikationen/73/diw_01.c. 575798.de/dp1700.pdf.
- [3] A. Stirling, P. Johnstone, A Global Picture of Industrial Interdependencies Between Civil and Nuclear Infrastructures. SPRU-Science and Technology Policy Research, Working Paper Series SWPS 2018-13 (August), University of Sussex, United Kingdom, 2018, http://dx.doi.org/10.2139/ssrn.3230021.
- [4] B. Wealer, S. Bauer, N. Landry, H. Seiß, C. von Hirschhausen, Nuclear Power Reactors Worldwide: Technology Developments, Diffusion Patterns, and Countryby-Country Analysis of Implementation (1951 - 2017). DIW Data Documentation, No. 93. DIW Berlin, Berlin, 2018, Retrieved from https://www.diw.de/documents/ publikationen/73/diw_01.c.583365.de/diw_datadoc_2018-093.pdf.
- [5] World Nuclear Association, Emerging Nuclear Energy Countries. http: //www.world-nuclear.org/information-library/country-profiles/others/ emerging-nuclear-energy-countries.aspx, 2019 (accessed 27 February 2019.
- [6] International Atomic Energy Agency, Energy, Electricity and Nuclear Power Estimates for the Period up to 2050, Reference Data Series No.1, 2018 Edition. International Atomic Energy Agency, Vienna, Austria, 2018, Retrieved from https://www-pub.iaea.org/MTCD/Publications/PDF/RDS-1-38_web.pdf.
- [7] J.M. Deutch, C.W. Forsberg, A.C. Kadak, M.S. Kazimi, E.J. Moniz, J.E. Parsons, Update of the MIT 2003 - Future of Nuclear Power. Massachusetts Institute of Technology, Cambridge, United States, 2009, Retrieved from http://web.mit.edu/ nuclearpower/pdf/nuclearpower-update2009.pdf.
- [8] L.W. Davis, Prospects for Nuclear Power, J. Econ. Perspect. 26(1) (2012) 49 66, https://doi.org/10.1257/jep.26.1.49.
- [9] F. Gralla, D.J. Abson, A.P. Møller, D.J. Lang, H. von Wehrden, Energy Transitions and National Development Indicators: A Global Review of Nuclear Energy Production, Renew. and Sustain. Energy Rev. 70 (2017) 1251 - 1265, https://doi.org/10.1016%2Fj.rser.2016.12.026.
- B.K. Sovacool, S.V. Valentine, The Socio-political Economy of Nuclear Energy in China and India, Energy 35(9) (2010) 3803 - 3813, https://doi.org/10.1016% 2Fj.energy.2010.05.033.
- [11] S.V. Valentine, B.K. Sovacool, The Socio-political Economy of Nuclear Power Development in Japan and South Korea, Energy Policy 38(12) (2010) 7971 - 7979, https://doi.org/10.1016%2Fj.enpol.2010.09.036.

- [12] PRIS Power Reactor Information System. https://pris.iaea.org/PRIS/home. aspx, 2019, (accessed 14 February 2019).
- [13] M. Fuhrmann, Splitting Atoms: Why do Countries build Nuclear Power Plants?, Int. Interact 38(1) (2012) 29 - 57, https://doi.org/10.1080%2F03050629.2012. 640209.
- [14] J. Jewell, S.A. Ates, Introducing Nuclear Power in Turkey: A Historic State Strategy and Future Prospects, Energy Res. Soc. Sci. 10 (2015) 273 - 282, https://doi.org/ 10.1016%2Fj.erss.2015.07.011.
- [15] F.W. Geels, J. Schot, Typology of Sociotechnical Transition Pathways, Res. Policy 36(3) (2007) 399 417, https://doi.org/10.1016/j.respol.2007.01.003.
- [16] F.W. Geels, Technological Transitions as Evolutionary Reconfiguration Processes: A Multi-Level Perspective and a Case-Study, Res. Policy 31(8-9) (2002) 1257 - 1274, https://doi.org/10.1016/s0048-7333(02)00062-8.
- [17] J. Markard, R. Raven, B. Truffer, Sustainability Transitions: An Emerging Field of Research and its Prospects, Res. Policy 41(6) (2002) 955 - 967, https://doi.org/ 10.1016/j.respol.2012.02.013.
- [18] N.E. Hultman, The Political Economy of Nuclear Energy, Wiley Interdisciplinary Reviews: Climate Change 2(3) (2011) 397 - 411, https://doi.org/10.1002%2Fwcc. 113.
- [19] A: Goldthau, Rethinking the Governance of Energy Infrastructure: Scale, Decentralization and Polycentrism, Energy Res. Soc. Sci. 1 (2014) 134 140, https://doi.org/10.1016/j.erss.2014.02.009.
- [20] P. Johnstone, A. Stirling, Comparing Nuclear Power Trajectories in Germany and the UK: From 'Regimes' to 'Democracies' in Sociotechnical Transitions and Discontinuities. SPRU-Science and Technology Policy Research, Working Paper Series SWPS 2015-18 (June), University of Sussex, United Kingdom, 2015, http: //dx.doi.org/10.2139/ssrn.2744549.
- [21] L. Diamond, L. Morlino, The Quality of Democracy: An Overview, J. Democr. 15(4) (2004) 20 31, https://doi.org/10.1353%2Fjod.2004.0060.
- [22] F: Zakaria, The Rise of Illiberal Democracy, Foreign Aff. 76(6) (1997) 22 43, https://doi.org/10.2307/20048274.
- [23] M.J. Burke, J.C. Stephens, Political Power and Renewable Energy Futures: A Critical Review, Energy Res. Soc. Sci. 35 (2018) 78 - 93, https://doi.org/10. 1016/j.erss.2017.10.018.
- [24] M. Bookchin, Towards an Ecological Society. Montreal: Black Rose Books, Montreal, Canada, 1996.
- [25] A. Lovins, Soft Energy Paths: Towards a Durable Peace. New York: Harper & Row, New York, United States, 1979.
- [26] H.P. Kitschelt, Political Opportunity Structures and Political Protest: Anti-nuclear Movements in four Democracies, Br. J. Political Sci., 16(1) (1986) 57 - 85, https: //doi.org/10.1017/s000712340000380x.

- [27] A.M. Weinberg, Views: The Maturity and Future of Nuclear Energy: The Most Serious Question Now Facing Nuclear Energy is its Acceptance by the Public, American Scientist, 64(1) (1976) 16 - 21, Retrieved from www.jstor.org/stable/ 27847037.
- [28] P.H. O'neil, Atoms and Democracy: Political Transition and the Role of Nuclear Energy, Democratization, 6(3) (1999) 171 - 189, https://doi.org/10.1080/ 13510349908403626.
- [29] J. Jewell, Ready for Nuclear Energy?: An Assessment of Capacities and Motivations for Launching New National Nuclear Power Programs, Energy Policy 39(3) (2011) 1041 - 1055, https://doi.org/10.1016/j.enpol.2010.10.041.
- [30] C.I. Barnard, J.R. Oppenheimer, C.A. Thomas, H.A. Winne, D.E. Lilienthal, A Report on the International Control of Atomic Energy. Prepared for the Secretary of State's Committee on Atomic Energy (known as: The Acheson-Lilienthal Report) Washington, D.C., United States, 1946, Retrieved from https://www.cia.gov/library/readingroom/docs/Report_on_ the_International_Control_of_Atomic_Energy_16_Mar_1946.PDF.
- [31] E. Yamamura, Effect of Free Media on Views Regarding Nuclear Energy After the Fukushima Accident, Kyklos 65(1) (2012) 132 - 141, https://doi.org/10.1111/j. 1467-6435.2011.00530.x.
- [32] M.G. Marshall, T.R. Gurr, K. Jaggers, POLITY IV PROJECT: Dataset Users' Manual. July 25, 2016 ed, 2017, Retrieved from https://www.systemicpeace.org/ inscr/p4manualv2016.pdf.
- [33] S. Haber, V. Menaldo, Do Natural Resources Fuel Authoritarianism? A Reappraisal of the Resource Curse, Am. Political Sci. Rev., 105(1) (2011) 1 - 26, https://doi. org/10.1017/s0003055410000584.
- [34] International Atomic Energy Agency, PRIS-STATISTICS Power Reactor Information System Statistical Reports User's Manual. COMPUTER MANUAL SERIES No. 22. International Atomic Energy Agency, Vienna, Austria, 2013, Retrieved from https://www-pub.iaea.org/MTCD/Publications/PDF/cms22_web.pdf.
- [35] J.M. Deutch, The New Nuclear Threat, Foreign Affairs 71(4) (1992) 120 134, https://doi.org/10.2307/20045313.
- [36] L. Mez, Nuclear Energy Any Solution for Sustainability and Climate Protection? Energy Policy, 48 (2012) 56 - 63, https://doi.org/10.1016/j.enpol.2012.04. 047.
- [37] H.M. Kristensen, R.S. Norris, Global Nuclear Weapons Inventories, 1945-2013, Bull. Atomic Scientists 69(5) (2013) 75 81, https://doi.org/10.1177/0096340213501363.
- [38] D.J. Jo, E. Gartzke, Determinants of Nuclear Weapons Proliferation, J. Confl. Resolut. 51(1) (2007) 167 - 194, https://doi.org/10.1177/0022002706296158.
- [39] P. Sadorsky, The Effect of Urbanization on CO₂ Emissions in Emerging Economies, Energy Econ. 41 (2014) 147 - 153, https://doi.org/10.1016/j.eneco.2013.11. 007.

- [40] B.K. Sovacool, Valuing the Greenhouse Gas Emissions from Nuclear Power: A Critical Survey, Energy Policy 36(8) (2008) 2950 - 2963, https://doi.org/10. 1016/j.enpol.2008.04.017.
- [41] M. Lenzen, Life Cycle Energy and Greenhouse Gas Emissions of Nuclear Energy: A review, Energy Convers. and Management 49(8) (2008) 2178 - 2199, https: //doi.org/10.1016/j.enconman.2008.01.033.
- [42] R. Turconi, A. Boldrin, T. Astrup, Life Cycle Assessment (LCA) of Electricity Generation Technologies: Overview, Comparability and Limitations, Renew. Sustain. Energy Rev., 28 (2013) 555 - 565, https://doi.org/10.1016/j.rser.2013.08. 013.
- [43] A. Verbruggen, E. Laes, S. Lemmens, Assessment of the Actual Sustainability of Nuclear Fission Power, Renew. Sustain. Energy Rev. 32 (2014) 16 - 28, https: //doi.org/10.1016/j.rser.2014.01.008.
- [44] R. Ingelhart, C. Welzel, Development and Democracy: What We know About Modernization Today. Foreign Affairs 5 (2009) 33 - 41, Retrieved from https: //ssrn.com/abstract=2391678.
- [45] M.T. Torfason, P. Ingram, The Global Rise of Democracy: A Network Account, American Sociological Rev. 75(3) (2010) 355 - 377, https://doi.org/10.1177/ 0003122410372230.
- [46] J. Hausman, D. McFadden, Specification Tests for the Multinomial Logit Model, Econometrica: Journal of the Econometric Society 52(5) (1984) 1219 - 1240, https: //doi.org/10.2307/1910997.
- [47] S.J. Long, J. Freese, Regression Models for Categorical Dependent Variables using Stata, Third Edition, Stata Press, Texas, United States, 2014, Retrieved from https://www.stata-press.com/books/regression-models-categoricaldependent-variables/.
- [48] S. Cheng, J.S. Long, Testing for IIA in the Multinomial Logit Model, Sociological Methods & Res. 35(4) (2007) 583 - 600, https://doi.org/10.1177/ 0049124106292361.
- [49] M. Hellström, I. Ruuska, K. Wikström, D. Jåfs, Project Governance and Path Creation in the early Stages of Finnish Nuclear Power Projects, Int. J. Proj. Management 31(5) (2013) 712 - 723, https://doi.org/10.1016/j.ijproman.2013.01.005.
- [50] R. Fouquet, 2016, Path Dependence in Energy Systems and Economic Development, Nature Energy (1) Article number: 16098, https://doi.org/10.1038%2Fnenergy. 2016.98.
- [51] D. Dassa Kaye, F.M. Wehrey, A nuclear Iran: The Reactions of Neighbours, Survival: Global Politics and Strategy 49(2) (2007) 111 - 128, https://doi.org/10.1080/ 00396330701437777

A Appendix

Variables	N	D1	W	GDP	U	E	EI	FFR	CO_2
N	1.000								
D1	0.351^{a}	1.000							
W	0.364^{a}	0.124^{a}	1.000						
GDP	0.243^{a}	0.314^{a}	0.116^{a}	1.000					
U	0.320^{a}	0.352^{a}	0.139^{a}	0.596^{a}	1.000				
E	0.313^{a}	0.280^{a}	0.122^{a}	0.812^{a}	0.583^{a}	1.000			
EI	0.104^{a}	0.213^{a}	0.042^{a}	-0.055^{a}	-0.024^{c}	0.011	1.000		
FFR	-0.087^{a}	-0.246^{a}	-0.027^{a}	0.060^{a}	0.187^{a}	0.068^{a}	-0.441^{a}	1.000	
CO_2	0.188^{a}	0.094^{a}	0.114^{a}	0.722^{a}	0.546^{a}	0.629^{a}	-0.196^{a}	0.312^{a}	1.000

 Table 10:
 Correlation matrix

Notes: Superscripts a, b, and c represent significance at 1%, 5%, and 10%, respectively.